Author’s response to reviews

Title: Transcriptional profiles of pilocytic astrocytoma are related to their three different locations, but not to radiological tumor features

Authors:

Krzysztof Zakrzewski (krzysztof.zakrzewski_xl@wp.pl)
Michal Jarzab (mjarzab@io.gliwice.pl)
Małgorzata Oczko-Wojciechowska (gosiaoczko@io.gliwice.pl)
Aleksandra Pfeifer (apfeifer@io.gliwice.pl)
Barbara Jarzab (bjarzab@io.gliwice.pl)
Pawel P Liberski (pawel.liberski@umed.lodz.pl)
Magdalena Zakrzewska (magdalena.zakrzewska@umed.lodz.pl)

Version: 6 Date: 12 October 2015

Author’s response to reviews: see over
Dear Editors,

Thank you very much for all comments. In the revised manuscript we included a new "Availability of Supporting Data" section with appropriate information about discussed data reposition. We also incorporate information concerning type of obtained consent within "Methods/Patient samples" section in accordance to Polish law.

I hope that the manuscript will be of interest for readers of BMC Cancer.

Yours faithfully,
Magdalena Zakrzewska